

## 

**Citation:** Toor J, Turner HC, Truscott JE, Werkman M, Phillips AE, Alsallaq R, et al. (2018) The design of schistosomiasis monitoring and evaluation programmes: The importance of collecting adult data to inform treatment strategies for *Schistosoma mansoni*. PLoS Negl Trop Dis 12(10): e0006717. https://doi.org/10.1371/journal. pntd.0006717

Editor: Clive Shiff, Johns Hopkins University, UNITED STATES

Received: March 20, 2018

Accepted: July 24, 2018

Published: October 8, 2018

**Copyright:** © 2018 Toor et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** This work was supported by the Bill and Melinda Gates Foundation [www.gatesfoundation. org] in partnership with the Task Force for Global Health through the NTD Modelling Consortium [grant number OPP1053230] to JT, RA, GFM, CHK and RMA, and through the Children's Investment RESEARCH ARTICLE

# The design of schistosomiasis monitoring and evaluation programmes: The importance of collecting adult data to inform treatment strategies for *Schistosoma mansoni*

Jaspreet Toor<sup>1\*</sup>, Hugo C. Turner<sup>2,3</sup>, James E. Truscott<sup>1,4</sup>, Marleen Werkman<sup>1,4</sup>, Anna E. Phillips<sup>1</sup>, Ramzi Alsallag<sup>5</sup>, Graham F. Medley<sup>6</sup>, Charles H. King<sup>5</sup>, Roy M. Anderson<sup>1,4</sup>

 London Centre for Neglected Tropical Disease Research and Department of Infectious Disease Epidemiology, School of Public Health, Faculty of Medicine, St Mary's Campus, Imperial College London, London, United Kingdom, 2 Oxford University Clinical Research Unit, Wellcome Trust Major Overseas Programme, Ho Chi Minh City, Vietnam, 3 Centre for Tropical Medicine and Global Health, Nuffield Department of Medicine, University of Oxford, Oxford, United Kingdom, 4 The DeWorm3 Project, The Natural History Museum of London, London, United Kingdom, 5 Center for Global Health and Diseases and Department of Mathematics, Case Western Reserve University, Cleveland, Ohio, United States of America, 6 Centre for Mathematical Modelling of Infectious Disease, London School of Hygiene and Tropical Medicine, London, United Kingdom

\* j.toor@imperial.ac.uk

## Abstract

Monitoring and evaluation (M&E) programmes are used to collect data which are required to assess the impact of current interventions on their progress towards achieving the World Health Organization (WHO) goals of morbidity control and elimination as a public health problem for schistosomiasis. Prevalence and intensity of infection data are typically collected from school-aged children (SAC) as they are relatively easy to sample and are thought to be most likely to be infected by schistosome parasites. However, adults are also likely to be infected. We use three different age-intensity profiles of infection for Schistosoma mansoni with low, moderate and high burdens of infection in adults to investigate how the age distribution of infection impacts the mathematical model generated recommendations of the preventive chemotherapy coverage levels required to achieve the WHO goals. We find that for moderate prevalence regions, regardless of the burden of infection in adults, treating SAC only may achieve the WHO goals. However, for high prevalence regions with a high burden of infection in adults, adult treatment is required to meet the WHO goals. Hence, we show that the optimal treatment strategy for a defined region requires consideration of the burden of infection in adults as it cannot be based solely on the prevalence of infection in SAC. Although past epidemiological data have informed mathematical models for the transmission and control of schistosome infections, more accurate and detailed data are required from M&E programmes to accurately determine the optimal treatment strategy for a defined region. We highlight the importance of collecting prevalence and intensity of infection data from a broader age-range, specifically the inclusion of adult data at baseline (prior to treatment) and throughout the treatment programme if possible, rather than SAC only, to accurately determine the treatment strategy for a defined region. Furthermore, we

Fund Foundation (UK) ("CIFF") [www.ciff.org] to JT and RMA. This work was also supported by the Bill and Melinda Gates Foundation [www.

gatesfoundation.org] for research grant support via the DeWorm3 [grant number OPP1129535] award to the Natural History Museum in London to JET, MW and RMA. Also, supported by the Wellcome Trust [grant number 089276/B/09/7; www. wellcome.ac.uk] to HCT. AEP was supported by the Medical Research Council (MRC) [www.mrc.ac.uk] through the Newton Fund. Additionally, supported by the Schistosomiasis Consortium for Operational Research (SCORE) [www.score.uga.edu] based at the University of Georgia to RA and CHK. The views, opinions, assumptions or any other information set out in this article are solely those of the authors and should not be attributed to the funders or any person connected with the funders. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** I have read the journal's policy and the authors of this manuscript have the following competing interests: RMA is a Non-Executive Director of GlaxoSmithKline (GSK). GSK played no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. The other authors have no competing interests.

discuss additional epidemiological data, such as individual longitudinal adherence to treatment, that should ideally be collected in M&E programmes.

### Author summary

Schistosomiasis remains an endemic parasitic disease affecting millions of people around the world. The World Health Organization (WHO) has set goals of morbidity control and elimination as a public health problem for schistosomiasis defined by reaching <5% and  $\leq$ 1% prevalence of heavy-intensity infections in school-aged children, respectively. Monitoring and evaluation (M&E) programmes are used to collect data which can inform treatment strategies required in a defined area and can also aid in assessing the progress of implemented treatment strategies. Due to programmatic and financial constraints, M&E data are typically collected from school-aged children as they are thought to be most likely to be infected. We highlight that adults should be included within M&E programmes by showing how the burden of infection in adults impacts our mathematical model recommendations of the treatment coverage levels required to reach the WHO goals for schistosomiasis. Our results highlight the importance of collecting data from a broader agerange, specifically the inclusion of adult data at baseline (prior to treatment) and throughout the treatment programme if possible. Improving M&E programmes to incorporate collection of such data will allow for more accurate determination of the optimal treatment strategy for a defined region.

## Introduction

Schistosomiasis is a neglected tropical disease (NTD) caused by parasitic worms which may infect the intestines (*Schistosoma mansoni* or *S. japonicum*) or urinary tract (*S. haematobium*). The disease is endemic in 54 countries affecting approximately 240 million people worldwide [1]. The World Health Organization (WHO) has set goals of morbidity control and elimination as a public health problem, i.e. reaching  $\leq$ 5% and  $\leq$ 1% prevalence of heavy-intensity infections in school-aged children (SAC; 5–14 years of age), respectively [2]. The overall aim is to achieve these goals using mass drug administration (MDA) of praziquantel at the treatment frequency and coverage level recommended by the WHO [1, 2]. The current WHO recommendations are treatment of SAC once every 3 years in low prevalence settings (<10% baseline prevalence among SAC), and treatment of SAC as well as adults considered to be at risk of infection once every 2 years in moderate prevalence settings (10–50% baseline prevalence among SAC) [1, 2]. Recent analyses based on mathematical models of schistosome transmission have recommended adaptations to these guidelines, such as an increase and expansion in treatment coverage levels to include adults or an increase in treatment frequency [3].

As SAC are most likely to be infected by *Schistosoma* parasites, possibly due to age-related water contact behaviour and/or the absence of acquired immunity generated by repeated exposure to infection, treatment has been specifically focused at this age group to prevent morbidity [4–6]. Given the high prevalence and intensity of infection typically in SAC, treating this age group may be more efficient and may have a greater impact on reducing transmission to the rest of the community. School-based treatment has been widely used as SAC are relatively easy to locate, sample and therefore reach good coverage levels. This makes SAC highly

important in terms of monitoring the impact of MDA treatment. By 2020, WHO aims to increase coverage such that 75% of SAC at risk will be regularly treated in endemic countries [7]. However, adults (ages 15+ as defined by WHO) are also likely to be infected [8, 9]. Hence, it is also important to include them in treatment and monitoring efforts since if MDA is only targeted at SAC, a large proportion of the local *Schistosoma* infection burden will remain untreated by chemotherapy creating a reservoir of infection within the community [8, 9]. In 2002, it was recommended that adults be treated in high-risk areas and that women of childbearing age not be excluded from MDA coverage [10]. However adult treatment has not been regularly implemented in most MDA programmes [1, 3, 8]. MDA coverage data collated by WHO reflect this situation [11]. Pre-school aged children (pre-SAC) may also be infected, though at present they are not eligible for treatment with praziquantel [12]. However, recent work shows that praziquantel may be used to treat pre-SAC provided the dosage is correct [13] so these guidelines may change in the coming years.

The WHO recognizes monitoring and evaluation (M&E) as an essential component of any treatment programme, and recommends the regular use of process and performance indicators for monitoring (i.e. assessing organizational elements and coverage levels), and performance and impact indicators for evaluation (i.e. assessing coverage levels, prevalence and intensity of infection) [1, 2]. It is vital that well-designed M&E activities are carried out to collect data that can assist in determining the progress of MDA programmes. This can give insight into the degree to which the prevalence and/or intensity of infection in a region has been reduced following treatment. Due to financial and programmatic constraints, data is typically collected from SAC only as they are most likely to be infected but also treated by national control programmes [14, 15]. Additionally, the preventive chemotherapy (PCT) strategy for a region is currently dependent upon the prevalence of infection in SAC at baseline (prior to treatment) with the WHO goals being dependent upon the prevalence of heavy-intensity infections in SAC only [1]. However, the prevalence of infection in SAC has been an inconsistent indicator of the burden of infection in adults [15–19]. The burden of infection in adults will vary geographically as it is likely to be impacted by local behavioural and cultural factors, as well as access to water, sanitation and hygiene (WASH) facilities [17].

The age profile of infection and more specifically, the burden of infection in adults relative to SAC, has been observed to vary for schistosome infections in different countries and communities [8, 17, 20–23]. In this study, we explore how different age profiles of infection with low to high burdens of infection in adults can impact our mathematical model generated recommendations of the treatment coverage levels required to achieve the WHO goals for *S. mansoni*. We show the importance of collecting broader age-specific data, particularly including adults as well as SAC, within M&E programmes. We also offer guidance on how M&E programmes should be designed by highlighting other data required to determine the progress of treatment strategies.

#### Methods

#### Model

We used an age-structured deterministic model of parasite transmission and control by praziquantel treatment developed by Imperial College London [8, 24] to follow through the currently recommended WHO guidelines for *S. mansoni* [2] (it is important to note that previous model analyses have suggested important adaptations to these guidelines [3]). Model parameters were based on previous studies (parameter values are shown in Table 1). To determine whether the burden of infection in adults relative to SAC would have an impact on the recommended treatment coverage levels, we used three different age-intensity profiles of infection

Parameter	Value	Source
Fecundity	0.34 eggs/female/sample	[27– 29]
Egg distribution within the individual	0.87	[ <u>28</u> , 29]
Aggregation parameter	0.04 (baseline SAC prevalence settings close to 10%); 0.24 (baseline SAC prevalence settings > 10%)	[ <u>8</u> , <u>30</u> ]
Density dependence fecundity	0.0007/female worm	[8, 17]
Worm lifespan	5.7 years	[21, 27]
Drug efficacy	86.3%	[31]
Low adult burden setting: Age specific contact rates for 0–5, 5–10, 10–16, 16+ years of age	0.01, 1.2, 1, 0.02	[17]
Moderate adult burden setting: Age specific contact rates for 0–5, 5–10, 10–16, 16+ years of age	0.032, 0.61, 1, 0.06	[17]
High adult burden setting: Age specific contact rates for 0–5, 5–12, 12–20, 20+ years of age	0.01, 0.61, 1, 0.12	[17]
Prevalence of infection	Percentage of population having egg count threshold (or eggs per gram, epg) $> 0$	-
Prevalence of heavy-intensity infections	Percentage of population having egg count threshold $\geq$ 16 (epg $\geq$ 400 divided by 24 to convert to egg count)	[32]
Human demography	Based on Uganda's demographic profile	[33, 34]

Table 1. Parameter values used within the age-structured deterministic model for Schistosoma mansoni (based	on
previous studies).	

https://doi.org/10.1371/journal.pntd.0006717.t001

which had a low, moderate or high burden of infection in the adult population. The age-intensity profiles were simulated in line with past epidemiological studies [21, 25, 26] and produced by varying the age-specific contact rates, i.e. the transmission intensity by age group, as shown in Fig 1 and Table 1 [17]. The prevalence in adults varied depending on the age-intensity profile of infection, i.e. as the burden of infection in adults relative to SAC increased, the peak of the age profile shifted to older age groups and the prevalence of infection in adults rose (baseline prevalence for adults shown in S1–S3 Tables). We then determined the coverage levels required to reach the WHO goals of morbidity control and elimination as a public health problem within a 5, 10 and 15-year programme.

#### Simulated treatment strategy

Beginning with an untreated population (run for 1000 years prior to initiating treatment in order to reach the endemic phase at a stable equilibrium), we implemented the frequency of PCT depending on the baseline prevalence in SAC (estimated by using Kato-Katz as the diagnostic test as currently recommended by WHO [2]). We carried out PCT as recommended by WHO at a 75% coverage level of SAC only whilst assuming 100% treatment adherence (with the drug efficacy of praziquantel set at 86.3% [31]). For moderate prevalence settings (10–50% SAC baseline prevalence), we carried out treatment once every 2 years and for high prevalence settings ( $\geq$ 50% SAC baseline prevalence), we carried out treatment once a year. To simulate a range of baseline prevalence levels falling within these settings, the intrinsic intensity of transmission, i.e. basic reproductive number ( $R_0$ ) [8], was varied in our model (such that, the prevalence at baseline increased with  $R_0$ ;  $R_0$  values shown in S1–S3 Tables). The aggregation



**Fig 1. Age-intensity profiles of infection.** Mean worm burden (eggs per gram of faeces, epg) across ages with low, moderate and high burdens of infection in adults relative to school-aged children (school-aged children are between 5–14 years of age and adults are 15+ years of age) [17].

https://doi.org/10.1371/journal.pntd.0006717.g001

parameter (k) was varied to define different prevalence levels at a stable equilibrium prior to treatment, i.e. lower baseline prevalence settings were reached using a lower k value reflecting higher degrees of parasite aggregation in the human host (k values shown in Table 1).

#### Model output

Following 5, 10 and 15 years of PCT, we determined the prevalence of heavy-intensity infections in SAC to investigate whether the WHO goals had been achieved (where the morbidity and elimination as a public health problem goals are reached once this prevalence is  $\leq$ 5% and  $\leq$ 1% in SAC, respectively). In cases where the WHO goals were not achievable within 5 to 15 years treatment of SAC only, we increased the level of SAC coverage and included adult treatment to determine the levels of coverage that would be required to make the goals achievable. Throughout the simulations of the deterministic model, we projected both the prevalence of infection (eggs per gram, epg > 0) and prevalence of heavy-intensity infections (epg  $\geq$  400; [1, 32]) in SAC and adults.

#### Results

In moderate prevalence settings, i.e. baseline prevalence 10–50% in SAC, we found that the WHO goals were likely to be achieved within 5 years following biennial treatment of 75% SAC only. This held regardless of whether the burden of infection in adults was low, moderate or high, as the WHO goals remained achievable with no adult treatment (Fig 2). However, for high prevalence settings (baseline prevalence  $\geq$ 50% in SAC), we found that the WHO goals were not likely to be achieved within 5 to 15 years following annual treatment with coverage at 75% of SAC only (Fig 3; aligning with previous work [3]). In these higher prevalence regions, treatment of adults as well as SAC was required within the treatment programme, with

Baseline prevalence in SAC	Morbidity control (≤5% heavy-intensity infections in SAC)	Elimination as a public health problem (<1% heavy-intensity infections in SAC)
Moderate (10-50%)	<u>5/10/15-year programme</u> SAC: 75%	<u>5/10/15-year programme</u> SAC: 75%
	<ul> <li>Low, moderate and high burden settings: Adults: 0%</li> </ul>	<ul> <li>Low, moderate and high burden settings: Adults: 0%</li> </ul>
	<u>5-year programme</u> SAC: 75%	<u>5-year programme</u> SAC: 75%
High (≥50%)	<ul> <li>Low burden setting: Adults: 0% - NA (SAC 89% + adults 75%)</li> <li>Moderate burden setting: Adults: 0% - NA (SAC 95% + adults 85%)</li> <li>High burden setting: Adults: 0% - NA (SAC 95% + adults 90%)</li> </ul>	<ul> <li>Low burden setting: Adults: 0% - NA (SAC 96% + adults 93%)</li> <li>Moderate burden setting: Adults: 0% - NA (SAC 100% + adults 100%)</li> <li>High burden setting: Adults: 0% - NA (SAC 100% + adults 100%)</li> </ul>
	10-year programme SAC: 75% • Low burden setting: Adults: 0% - 3% • Moderate burden setting: Adults: 0% - 86% (or SAC 80% + adults 43%) • High burden setting: Adults: 0% - 77% (or SAC 85% + adults 63%)	<u>10-year programme</u> SAC: 75% • Low burden setting: Adults: 0% - 86% (or SAC 80% + adults 16%) • Moderate burden setting: Adults: 0% - NA (SAC 85% + adults 52%) • High burden setting: Adults: 0% - 94% (or SAC 85% + adults 76%)
	<u>15-year programme</u> SAC: 75% • Low burden setting: Adults: 0% • Moderate burden setting: Adults: 0% - 25% • High burden setting: Adults: 0% - 60%	<u>15-year programme</u> SAC: 75% • Low burden setting: Adults: 0% • Moderate burden setting: Adults: 0% - 50% • High burden setting: Adults: 0% - 70% (or SAC 80% + adults 63%)

Fig 2. Coverage levels required to meet the World Health Organization (WHO) goals. Levels of school-aged children (SAC; 5–14 years of age) and adult ( $\geq$  15 years of age) coverage required to meet the WHO goals when following currently recommended WHO treatment frequencies, i.e. for moderate baseline prevalence in SAC, treating once every 2 years; for high baseline prevalence in SAC, treating once a year. Required coverage levels are shown for a 5, 10 and 15-year treatment programme within low, moderate and high adult burden of infection settings. NA: not achievable unless SAC coverage is increased above 75%. More detailed S1–S3 Tables.

https://doi.org/10.1371/journal.pntd.0006717.g002

coverage levels varying with the burden of infection in adults (Fig 4; though for baseline prevalence levels close to 50% in SAC, the goals were achievable with 75% SAC only treatment).

Given a high burden of infection in adults, the baseline prevalence of infection and heavyintensity infections in adults was higher than SAC (Fig 3). Following treatment of SAC only, the prevalence of infection and heavy-intensity infections in SAC declined but remained high in adults, regardless of the burden of infection in adults (Fig 3). When not included within the treatment programme, infected adults remained as a reservoir of infection, making it vital that they are included within treatment. Expanding to include adult treatment led to decreased prevalence of infection and heavy-intensity infections in adults and simultaneously had a beneficial impact on further reducing the prevalence in SAC relative to treating SAC only (Figs 3 and 4).

For high prevalence settings, the SAC and adult coverage levels required to reach the WHO goals varied depending on the burden of infection in adults used within the age-intensity profile of infection. Given a low burden of infection in adults, 89% SAC and 75% adults need to be treated to achieve the morbidity goal within 5 years, and 75% SAC only need to be treated to achieve it within 15 years. Given a moderate burden of infection in adults, 95% SAC and 85% adults need to be treated to achieve the morbidity goal within 5 years, and 75% SAC and 25% adults need to be treated to achieve it within 15 years. Given a moderate burden of infection in adults, 95% SAC and 25% adults need to be treated to achieve it within 15 years. Given a high burden of infection in



Fig 3. Model projections of annual treatment of 75% school-aged children (SAC; 5–14 years of age) and 0% adults (15+ years of age). Shown for high prevalence settings ( $\geq$ 50% SAC baseline prevalence) for (a) low (R<sub>0</sub> = 3.0), (b) moderate (R<sub>0</sub> = 3.5) and (c) high (R<sub>0</sub> = 4.0) adult burdens of infection.

https://doi.org/10.1371/journal.pntd.0006717.g003

adults, 95% of SAC and 90% of adults need to be treated to achieve the morbidity goal within 5 years, and 75% SAC and 60% adults need to be treated to achieve it within 15 years (Figs 2, 5 and S1A).

The coverage levels increased when the target was the WHO goal of elimination as a public health problem. Given a low burden of infection in adults, 96% SAC and 93% adults need to be treated to achieve the elimination goal within 5 years, and 75% SAC only need to be treated to achieve it within 15 years. Given a moderate burden of infection in adults, all SAC and adults need to be treated to achieve the elimination goal within 5 years, and 75% SAC and 50% adults need to be treated to achieve it within 15 years. Given a high burden of infection in adults, all SAC and adults need to be treated to achieve the elimination goal within 5 years, and 75% SAC and 70% adults need to be treated to achieve it within 15 years (Figs 2, 6 and S1B). Overall, the level of adult coverage required to reach the WHO goals increased as the burden of infection in adults increased (Figs 4–6 and S1). As the number of years of PCT taking place increased, the coverage levels required to achieve the WHO goals decreased as expected.

In summary, the model recommendations for PCT coverage levels vary depending on the age-intensity profile of the region. We have highlighted that the burden of infection in adults as well as SAC will aid in informing the optimal treatment strategy for a region. Hence prevalence and intensity of infection data needs to be collected from adults, in addition to SAC, to inform their burden of infection, both at baseline and during regular surveillance to monitor the treatment programme. More detailed results are shown in <u>S1–S3</u> Tables.



Fig 4. Model projections of annual treatment for high prevalence settings ( $\geq$ 50% SAC baseline prevalence) using coverage levels required to achieve elimination as a public health problem within 10 years. With (a) low adult burden of infection ( $R_0 = 3.0$ ) treating 80% school-aged children (SAC; 5–14 years of age) and 16% adults (15+ years of age), (b) moderate adult burden of infection ( $R_0 = 3.5$ ) treating 85% SAC and 52% adults, and (c) high adult burden of infection ( $R_0 = 4.0$ ) treating 85% SAC and 76% adults.

https://doi.org/10.1371/journal.pntd.0006717.g004

#### Discussion

In this study, we have investigated the impact of the burden of infection in adults on the coverage levels that are required to reach the WHO goals for *S. mansoni*. The levels of coverage vary depending on the transmission setting (i.e. R<sub>0</sub>; aligning with previous work [3]). In moderate baseline prevalence settings, the WHO goals of morbidity control and elimination as a public health problem are likely to be achieved following treatment of SAC only, regardless of the



Fig 5. Coverage levels required to reach the World Health Organization (WHO) goal of morbidity control. High prevalence settings ( $\geq$ 50% SAC baseline prevalence) showing coverage levels of school-aged children (SAC; 5–14 years of age) and adults required to reach the WHO goal of morbidity control within 5- and 10-year treatment programmes (see <u>S1A Fig</u> for 15-year treatment programme).

https://doi.org/10.1371/journal.pntd.0006717.g005

High adult burden

Moderate adult burden



Low adult burden



https://doi.org/10.1371/journal.pntd.0006717.g006

burden of infection in adults. However, in high baseline prevalence settings, the untreated reservoir of infection in adults can maintain infection within the whole community making it clear that adult data needs to be collected to inform the optimal treatment coverage levels. Although the prevalence of infection in SAC can be predictive of the burden of infection in adults in some settings, this has been an inconsistent indicator [15–19]. Our results have shown that it is vital that adults are included in M&E programmes, particularly at baseline, as this will inform the optimal PCT strategy for a defined region. We have highlighted additional data requirements in Table 2.

One limitation of this study is that there is currently a lack of data on individual adherence following repeated treatment. At present, no recommendations are made about monitoring adherence to treatment at each round of MDA. Within our model, we have assumed that coverage at each round of treatment is at random, i.e. the individuals receiving treatment in each round are chosen at random. However, if there are persistent non-adherers to treatment (or subgroups such as pregnant women that are often excluded from national treatment programmes), then higher and/or broader age-related coverage levels will likely be required to reach the WHO goals. Hence, collecting longitudinal data on adherence to treatment by individuals within M&E programmes is vital. Another limitation is that using these parameter values, our model is unable to reach a stable equilibrium for low prevalence settings (<10% SAC baseline prevalence). Additionally, we are assuming that the age-intensity profile of infection is formed by the same mechanism following treatment. However, if some of the reduction in prevalence and/or intensity of infection in adults is due to acquired immunity (which may be more relevant when extending this analysis to S. haematobium), then treatment will also impact on the underlying mechanism that creates the age profile. For example, under the assumptions that acquired immunity is an important determinant of infection in older individuals and treatment of SAC only, this age group may then be less immune as adults, leading to a rise in adult prevalence over time as MDA impact on transmission increases. It is likely that adult treatment would be important and required to prevent this but only knowledge of the dynamics of burdens in adults can reveal this requirement. Consequently, we recommend regular assessment of prevalence and intensity of infection in adults, particularly in cases where the prevalence in SAC does not decline following treatment.

There are logistical difficulties in collecting data from adults. SAC are typically targeted by control programmes as they are relatively easy to reach within school settings and therefore coverage could be high. Some past field studies suggest that treating adults has required more personnel, time, and resources to find and obtain follow-up samples [18]. We note these

Baseline data	<ul> <li>Demographic census data on sub-populations (adults, SAC attending school, total SAC) required as denominators in assessing prevalence and treatment coverage [35].</li> <li>Mapping data of the region given the spatial heterogeneity of the disease [36].</li> <li>Data from low prevalence areas as we move towards elimination since most available data are from higher prevalence areas.</li> <li>Data sharing among investigators.</li> </ul>
Prevalence and intensity of infection data	<ul> <li>At baseline and following treatment, cross-sectional and longitudinal infection intensity and prevalence data across all age classes to indicate disease status, produce pre- and post-treatment age profiles of infection and estimate R<sub>0</sub> [8].</li> <li>Following treatment, prevalence of infection may bounce-back but likely at a lower average intensity of infection [37].</li> <li>Longitudinal records of egg counts with corresponding data on age and treatment history to record individual intensities of infection over time [24, 38].</li> </ul>
Treatment data	• Treatment coverage and adherence data at an individual level at each round of treatment to estimate coverage and adherence by age group to determine whether the target coverage level has been achieved [8, 24, 38].
Water, sanitation and hygiene (WASH) data	<ul> <li>Another NTD intervention is prevention of infection by improving WASH measures. WASH interventions can be difficult to distinguish from MDA since both are likely to have a similar impact on the prevalence of infection.</li> <li>Monitoring of facilities to check whether they have been purchased and installed. Individual and/or house-hold behaviour monitored to see if the facilities are being used (can be difficult to measure as behavioural regression may arise).</li> </ul>
Snail data	Snail abundance and infection data is required as this is currently lacking.
Parasite data	<ul> <li>Parasite life expectancy data in humans as this can range from 3.5 to 10.5 years for schistosomiasis worms [8, 21, 27].</li> <li>Data on the nature of density-dependent fecundity [8].</li> <li>Autopsy data on parasite loads stratified by sex in infected patients [39].</li> </ul>
Diagnostic techniques and cost data	<ul> <li>Kato-Katz needed to compare to baseline data, unless CCA can be related to Kato-Katz.</li> <li>Cost data as costs of the diagnostic tests vary in different settings and costs per test are not constant [14, 40].</li> <li>Data on which age groups are infected to assist in determining the most cost-effective treatment strategy [8, 14, 17].</li> </ul>

#### Table 2. Data requirements for monitoring and evaluation (M&E) programmes.

https://doi.org/10.1371/journal.pntd.0006717.t002

challenges but recommend collecting adult data, at least at baseline (prior to treatment), to inform PCT and then following each round of treatment if possible. Such challenges could be reduced following effective sensitization within a community. Importantly, the adults should be sampled at random to avoid misleading results. The degree of infection in the adult population and the peak age of infection are key features to capture as the age-profile of infection at baseline will assist in determining which age group to treat for schistosomiasis infection. Such baseline data from a wider range of ages in the population will permit more precise parameterization of the model and hence better predictions of the levels of treatment required to achieve the WHO goals. By not treating infected individuals (including any infected persistent nonadherers to treatment) a reservoir of infection will remain [8, 9] making the WHO goals more difficult to achieve. We have shown that treating SAC only can leave adults heavily infected. Furthermore, it is important that both prevalence and intensity of infection data is collected from adults as well as SAC. The relationship between prevalence and mean intensity of infection is non-linear with the precise relationship depending on the degree of parasite aggregation in the human host population as measured inversely by the negative binomial aggregation parameter (k) [6, 41-43]. Following treatment, prevalence typically bounces-back but at a lower intensity [37].

Diagnostic techniques are becoming increasingly important as prevalence needs to be accurately detected at low levels, particularly as programmes move from morbidity control towards elimination. Hence, it is important to consider which diagnostic techniques will be used in monitoring schistosomiasis infection. The most common diagnostic tools for schistosomiasis are the traditional Kato-Katz diagnostic technique (which current WHO guidelines are based on [2]) and the relatively newer, more sensitive point-of-care circulating cathodic antigen test (CCA) diagnostic test [37, 44, 45]. Although Kato-Katz is seen as the cheaper test, given the increased sensitivity of CCA, this may outweigh costs in long term as the CCA test is faster, less labour intensive and likely to have a higher prevalence threshold [14, 37, 46]. However, the costs vary in different settings and costs per test are not constant [14]. Additionally, there are no current WHO guidelines on mapping with such new diagnostics, nor subsequent treatment cut-offs using more sensitive tools. Economies of scale are important when considering the costs of MDA as this means the cost per treatment decreases as the number of people treated increases [40]. Currently, programmes focus on sampling SAC due to financial and programmatic constraints [14]. Accurate data on which age groups are infected are required to assess for the most cost-effective treatment strategy [14]. Hence, the need for adult inclusion within M&E programmes highlights the need for more cost-effective, rapid and accurate diagnostics [14, 17]. Notably, there is a need to obtain data relating diagnostic techniques (such as Kato Katz and CCA on samples from the same individual at the same time) and more accurate cost data.

We have also highlighted other data that should be collected within M&E programmes, including more accurate, detailed prevalence and intensity of infection data, treatment coverage and adherence data, and water, sanitation and hygiene (WASH) data (Table 2). By designing M&E programmes to collect this information, our models can be better parameterized to determine the projected impact of treatment in an area. Furthermore, with better M&E programmes, we can more accurately assess the progress of treatment programmes and improve recommendations on optimal treatment strategies. Well-documented M&E programmes will allow us to assess PCT strategies and aid in the development of additional improvements.

## Conclusions

For moderate prevalence settings, treating school-aged children only may achieve the WHO goals, regardless of the burden of infection in adults. However, for high prevalence settings, treatment of adults as well as SAC is required within the treatment programme, with coverage levels varying with the burden of infection in adults. Hence, the optimal treatment strategy for a defined region cannot be determined solely by the level of infection in SAC. This highlights that prevalence and intensity of infection data in adults needs to be included within M&E programmes, particularly at baseline but also following treatment when monitoring the impact of treatment. The burden of infection in adults will better inform whether adult treatment is required and the coverage levels which are required to achieve the WHO goals and elimination as we look further to the future.

## **Supporting information**

**S1 Fig. Coverage levels required to reach the World Health Organization (WHO) goals.** High baseline prevalence settings (≥50% SAC baseline prevalence) showing coverage levels of school-aged children (SAC; 5–14 years of age) and adults required to reach the WHO goals of (a) morbidity control and (b) elimination as a public health problem within a 15-year treatment programme. (TIF) S1 Table. Low burden setting in adults. Levels of school-aged children (SAC; 5–14 years of age) and adult ( $\geq$  15 years of age) coverage required to meet the WHO goals when following currently recommended WHO treatment frequencies, i.e. for moderate baseline prevalence in SAC, treating once every 2 years; for high baseline prevalence in SAC, treating once a year. Required coverage levels are shown for a 5, 10 and 15-year treatment programme. NA: not achievable unless SAC coverage is increased above 75%. (DOCX)

S2 Table. Moderate burden setting in adults. Levels of school-aged children (SAC; 5–14 years of age) and adult ( $\geq$  15 years of age) coverage required to meet the WHO goals when following currently recommended WHO treatment frequencies, i.e. for moderate baseline prevalence in SAC, treating once every 2 years; for high baseline prevalence in SAC, treating once a year. Required coverage levels are shown for a 5, 10 and 15-year treatment programme. NA: not achievable unless SAC coverage is increased above 75%. (DOCX)

S3 Table. High burden setting in adults. Levels of school-aged children (SAC; 5–14 years of age) and adult ( $\geq$  15 years of age) coverage required to meet the WHO goals when following currently recommended WHO treatment frequencies, i.e. for moderate baseline prevalence in SAC, treating once every 2 years; for high baseline prevalence in SAC, treating once a year. Required coverage levels are shown for a 5, 10 and 15-year treatment programme. NA: not achievable unless SAC coverage is increased above 75%. (DOCX)

### Acknowledgments

We thank Deirdre Hollingsworth and Simon Brooker for helpful comments and discussions on the study. Additionally, we thank Hajnal Farkas for helping to manage the NTD Modelling Consortium.

#### **Author Contributions**

**Conceptualization:** Jaspreet Toor, Hugo C. Turner, James E. Truscott, Marleen Werkman, Charles H. King, Roy M. Anderson.

Formal analysis: Jaspreet Toor.

Investigation: Jaspreet Toor.

Methodology: Jaspreet Toor, Hugo C. Turner, James E. Truscott.

Writing - original draft: Jaspreet Toor.

Writing – review & editing: Jaspreet Toor, Hugo C. Turner, Marleen Werkman, Anna E. Phillips, Ramzi Alsallaq, Graham F. Medley, Charles H. King, Roy M. Anderson.

#### References

- 1. World Health Organization. Schistosomiasis: Progress report 2001–2011 and strategic plan 2012–2020. 2013.
- 2. World Health Organization. Helminth control in school-age children: A guide for managers of control programmes. 2011.

- Toor J, Alsallaq RA, Truscott JE, Turner HC, Werkman M, Gurarie D, et al. Are we on our way to achieving the 2020 goals for schistosomiasis morbidity control using current WHO guidelines? Clinical Infectious Diseases. 2018. https://doi.org/10.1093/cid/ciy001 PMID: 29860290
- Butterworth AE, Dalton PR, Dunne DW, Mugambi M, Ouma JH, Richardson BA, et al. Immunity after treatment of human schistosomiasis mansoni. I. Study design, pretreatment observations and the results of treatment. Trans R Soc Trop Med Hyg. 1984; 78(1):108–23. PMID: 6710563.
- Kabatereine NB, Vennervald BJ, Ouma JH, Kemijumbi J, Butterworth AE, Dunne DW, et al. Adult resistance to schistosomiasis mansoni: age-dependence of reinfection remains constant in communities with diverse exposure patterns. Parasitology. 1999; 118 (Pt 1):101–5. PMID: 10070667.
- Kabatereine NB, Brooker S, Tukahebwa EM, Kazibwe F, Onapa AW. Epidemiology and geography of Schistosoma mansoni in Uganda: implications for planning control. Trop Med Int Health. 2004; 9 (3):372–80. PMID: 14996367.
- World Health Assembly. Schistosomiasis and soil-transmitted helminth infections. WHA resolution 54.19. 2001.
- Anderson RM, Turner HC, Farrell SH, Truscott JE. Studies of the transmission dynamics, mathematical model development and the control of schistosome parasites by mass drug administration in human communities. Adv Parasitol. 2016; 94:199–246. https://doi.org/10.1016/bs.apar.2016.06.003 PMID: 27756455.
- Njenga SM, Mwandawiro CS, Muniu E, Mwanje MT, Haji FM, Bockarie MJ. Adult population as potential reservoir of NTD infections in rural villages of Kwale district, Coastal Kenya: implications for preventive chemotherapy interventions policy. Parasit Vectors. 2011; 4:175. https://doi.org/10.1186/1756-3305-4-175 PMID: 21917166; PubMed Central PMCID: PMCPMC3212820.
- World Health Organization. Report of the WHO informal consultation on the use of praziquantel during pregnancy/lactation and albendazole/mebendazole in children under 24 months. 2003.
- World Health Organization. PCT databank: Schistosomiasis 2017. Available from: <a href="http://www.who.int/neglected\_diseases/preventive\_chemotherapy/sch/en/">http://www.who.int/neglected\_diseases/preventive\_chemotherapy/sch/en/</a>.
- World Health Organization. Preventive chemotherapy in human helminthiasis: coordinated use of anthelminthic drugs in control interventions: a manual for health professionals and programme managers. 2006.
- Montresor A, Garba A. Treatment of preschool children for schistosomiasis. Lancet Glob Health. 2017; 5(7):e640–e1. https://doi.org/10.1016/S2214-109X(17)30202-4 PMID: 28619212.
- Turner HC, Bettis AA, Dunn JC, Whitton JM, Hollingsworth TD, Fleming FM, et al. Economic Considerations for Moving beyond the Kato-Katz Technique for Diagnosing Intestinal Parasites As We Move Towards Elimination. Trends Parasitol. 2017; 33(6):435–43. https://doi.org/10.1016/j.pt.2017.01.007 PMID: 28187989; PubMed Central PMCID: PMCPMC5446322.
- Guyatt HL, Brooker S, Donnelly CA. Can prevalence of infection in school-aged children be used as an index for assessing community prevalence? Parasitology. 1999; 118 (Pt 3):257–68. PMID: 10205801.
- Evans DS, King JD, Eigege A, Umaru J, Adamani W, Alphonsus K, et al. Assessing the WHO 50% prevalence threshold in school-aged children as indication for treatment of urogenital schistosomiasis in adults in central Nigeria. Am J Trop Med Hyg. 2013; 88(3):441–5. https://doi.org/10.4269/ajtmh.12-0511 PMID: 23382170; PubMed Central PMCID: PMCPMC3592522.
- Turner HC, Truscott JE, Bettis AA, Farrell SH, Deol AK, Whitton JM, et al. Evaluating the variation in the projected benefit of community-wide mass treatment for schistosomiasis: Implications for future economic evaluations. https://doi.org/10.1186/s13071-017-2141-5 PMID: 28454578. 2017; 10(1):213. PubMed Central PMCID: PMCPMC5410074.
- Mwinzi PN, Muchiri G, Wiegand RE, Omedo M, Abudho B, Karanja DM, et al. Predictive Value of School-Aged Children's Schistosomiasis Prevalence and Egg Intensity for Other Age Groups in Western Kenya. Am J Trop Med Hyg. 2015; 93(6):1311–7. https://doi.org/10.4269/ajtmh.15-0467 PMID: 26416108; PubMed Central PMCID: PMCPMC4674251.
- Pereira AP, Favre TC, Galvao AF, Beck L, Barbosa CS, Pieri OS. The prevalence of schistosomiasis in school-aged children as an appropriate indicator of its prevalence in the community. Mem Inst Oswaldo Cruz. 2010; 105(4):563–9. PMID: 20721509.
- McCullough FS, Bradley DJ. Egg output stability and the epidemiology of Schistosoma haematobium. I. Variation and stability in Schistosoma haematobium egg counts. Trans R Soc Trop Med Hyg. 1973; 67 (4):475–90. PMID: 4785458.
- Fulford AJ, Butterworth AE, Ouma JH, Sturrock RF. A statistical approach to schistosome population dynamics and estimation of the life-span of *Schistosoma mansoni* in man. Parasitology. 1995; 110 (Pt 3):307–16. PMID: 7724238.

- Butterworth AE, Sturrock RF, Ouma JH, Mbugua GG, Fulford AJ, Kariuki HC, et al. Comparison of different chemotherapy strategies against Schistosoma mansoni in Machakos District, Kenya: effects on human infection and morbidity. Parasitology. 1991; 103 Pt 3:339–55. PMID: 1780171.
- Fulford AJ, Butterworth AE, Sturrock RF, Ouma JH. On the use of age-intensity data to detect immunity to parasitic infections, with special reference to Schistosoma mansoni in Kenya. Parasitology. 1992; 105 (Pt 2):219–27. PMID: 1454421.
- Truscott JE, Gurarie D, Alsallaq R, Toor J, Yoon N, Farrell SH, et al. A comparison of two mathematical models of the impact of mass drug administration on the transmission and control of schistosomiasis. Epidemics. 2017; 18:29–37. https://doi.org/10.1016/j.epidem.2017.02.003 PMID: 28279453; PubMed Central PMCID: PMCPMC5340850.
- 25. King CH, Magak P, Salam EA, Ouma JH, Kariuki HC, Blanton RE, et al. Measuring morbidity in schistosomiasis mansoni: relationship between image pattern, portal vein diameter and portal branch thickness in large-scale surveys using new WHO coding guidelines for ultrasound in schistosomiasis. Trop Med Int Health. 2003; 8(2):109–17. PMID: 12581434.
- DeStigter KV, King CH, Keating CE, Ouma JH, Siongok TK, Mahmoud AA. Effect of targeted mass treatment on intensity of infection and morbidity in schistosomiasis mansoni: seven-year follow-up of a community in Machakos, Kenya. Trans Assoc Am Physicians. 1989; 102:209–12. PMID: 2517855.
- Anderson RM, May RM. Population dynamics of human helminth infections: control by chemotherapy. Nature. 1982; 297(5867):557–63. PMID: 7088139.
- De Vlas SJ, Gryseels B, Van Oortmarssen GJ, Polderman AM, Habbema JD. A model for variations in single and repeated egg counts in Schistosoma mansoni infections. Parasitology. 1992; 104 (Pt 3):451–60. PMID: 1641245.
- de Vlas SJ, Nagelkerke NJ, Habbema JD, van Oortmarssen GJ. Statistical models for estimating prevalence and incidence of parasitic diseases. Stat Methods Med Res. 1993; 2(1):3–21. https://doi.org/10. 1177/096228029300200102 PMID: 8261249.
- Chan MS, Guyatt HL, Bundy DA, Booth M, Fulford AJ, Medley GF. The development of an age structured model for schistosomiasis transmission dynamics and control and its validation for Schistosoma mansoni. Epidemiol Infect. 1995; 115(2):325–44. PMID: 7589272; PubMed Central PMCID: PMCPMC2271402.
- Zwang J, Olliaro PL. <a href="https://doi.org/10.1371/journal.pntd.0003286">https://doi.org/10.1371/journal.pntd.0003286</a> PMID: 25412105. PLoS Negl Trop Dis. 2014; 8(11):e3286. PubMed Central PMCID: PMCPMC4238982.
- World Health Organization Expert Committee. Prevention and control of schistosomiasis and soil-transmitted helminthiasis. World Health Organization Techical Report Series. 2002; 912:i–vi, 1–57, back cover. PMID: 12592987.
- Anderson R, Truscott J, Hollingsworth TD. The coverage and frequency of mass drug administration required to eliminate persistent transmission of soil-transmitted helminths. Philos Trans R Soc Lond B Biol Sci. 2014; 369(1645):20130435. <u>https://doi.org/10.1098/rstb.2013.0435</u> PMID: 24821921; PubMed Central PMCID: PMCPMC4024228.
- Pullan RL, Kabatereine NB, Quinnell RJ, Brooker S. Spatial and genetic epidemiology of hookworm in a rural community in Uganda. PLoS Negl Trop Dis. 2010; 4(6):e713. https://doi.org/10.1371/journal.pntd. 0000713 PMID: 20559556; PubMed Central PMCID: PMCPMC2886101.
- Shuford KV, Turner HC, Anderson RM. Compliance with anthelmintic treatment in the neglected tropical diseases control programmes: a systematic review. Parasit Vectors. 2016; 9:29. https://doi.org/10. 1186/s13071-016-1311-1 PMID: 26813098; PubMed Central PMCID: PMCPMC4729159.
- Kittur N, Binder S, Campbell CH, King CH, Kinung'hi S, Olsen A, et al. Defining Persistent Hotspots: Areas That Fail to Decrease Meaningfully in Prevalence after Multiple Years of Mass Drug Administration with Praziquantel for Control of Schistosomiasis. Am J Trop Med Hyg. 2017; 97(6):1810–7. https:// doi.org/10.4269/ajtmh.17-0368 PMID: 29016344.
- Prada JM, Touloupou P, Adriko M, Tukahebwa EM, Lamberton PHL, Hollingsworth TD. Understanding the relationship between egg- and antigen-based diagnostics of Schistosoma mansoni infection preand post-treatment in Uganda. Parasit Vectors. 2018; 11(1):21. https://doi.org/10.1186/s13071-017-2580-z PMID: 29310695; PubMed Central PMCID: PMCPMC5759883.
- Anderson RM, Turner HC, Farrell SH, Yang J, Truscott JE. What is required in terms of mass drug administration to interrupt the transmission of schistosome parasites in regions of endemic infection? Parasit Vectors. 2015; 8:553. https://doi.org/10.1186/s13071-015-1157-y PMID: 26489831; PubMed Central PMCID: PMCPMC4618750.
- Cheever AW, Andrade ZA. Pathological lesions associated with Schistosoma mansoni infection in man. Trans R Soc Trop Med Hyg. 1967; 61(5):626–39. PMID: 4293432.

- Turner HC, Toor J, Hollingsworth TD, Anderson RM. Economic evaluations of mass drug administration: The importance of economies of scale and scope. Clin Infect Dis. 2017. https://doi.org/10.1093/ cid/cix1001 PMID: 29126255.
- Anderson RM, May RM. Herd immunity to helminth infection and implications for parasite control. Nature. 1985; 315(6019):493–6. PMID: 4000277.
- Guyatt HL, Bundy DA. Estimating prevalence of community morbidity due to intestinal helminths: prevalence of infection as an indicator of the prevalence of disease. Trans R Soc Trop Med Hyg. 1991; 85 (6):778–82. PMID: 1801353.
- Guyatt HL, Smith T, Gryseels B, Lengeler C, Mshinda H, Siziya S, et al. Aggregation in schistosomiasis: comparison of the relationships between prevalence and intensity in different endemic areas. Parasitology. 1994; 109 (Pt 1):45–55. PMID: 8058368.
- 44. Lamberton PH, Kabatereine NB, Oguttu DW, Fenwick A, Webster JP. Sensitivity and specificity of multiple Kato-Katz thick smears and a circulating cathodic antigen test for Schistosoma mansoni diagnosis pre- and post-repeated-praziquantel treatment. PLoS Negl Trop Dis. 2014; 8(9):e3139. https://doi.org/ 10.1371/journal.pntd.0003139 PMID: 25211217; PubMed Central PMCID: PMCPMC4161328.
- 45. Kittur N, Castleman JD, Campbell CH Jr., King CH, Colley DG. Comparison of Schistosoma mansoni Prevalence and Intensity of Infection, as Determined by the Circulating Cathodic Antigen Urine Assay or by the Kato-Katz Fecal Assay: A Systematic Review. Am J Trop Med Hyg. 2016; 94(3):605–10. https://doi.org/10.4269/ajtmh.15-0725 PMID: 26755565; PubMed Central PMCID: PMCPMC4775897.
- 46. Werkman M, Truscott JE, Toor J, Wright JE, Anderson RM. The past matters: estimating intrinsic hookworm transmission intensity in areas with past mass drug administration to control lymphatic filariasis. Parasit Vectors. 2017; 10(1):254. https://doi.org/10.1186/s13071-017-2177-6 PMID: 28535806; PubMed Central PMCID: PMCPMC5493118.